**Author’s response to reviews**

**Title:** Patient and Provider experience and perspectives of a risk-based approach to multidisciplinary chronic kidney disease care: a mixed methods study

**Authors:**

Michelle Smekal (msmekal@ucalgary.ca)

Helen Tam-Tham (tamh@ucalgary.ca)

Juli Finlay (jlfinlay@ucalgary.ca)

Maoliosa Donald (donaldm@ucalgary.ca)

Chandra Thomas (Chandra.Thomas@albertahealthservices.ca)

Robert Weaver (rgweaver@ucalgary.ca)

Robert Quinn (rob.quinn@albertahealthservices.ca)

Kin Tam (kin.tam@ahs.ca)

Braden Manns (Braden.Manns@albertahealthservices.ca)

Marcello Tonelli (cello@ucalgary.ca)

Aminu Bello (aminu1@ualberta.ca)

Navdeep Tangri (ntangri@sogh.mb.ca)

Brenda Hemmelgarn (Brenda.Hemmelgarn@ahs.ca)

**Version:** 4 **Date:** 15 Feb 2019

**Author’s response to reviews:**

Dear Dr Zhang

We have completed the requests:

1) emails changed
2) abbreviations added
3) figures are currently in correct order
Kind regards

Brenda